Author's response to reviews

Title: Study flow diagrams in Cochrane systematic review updates. An adapted PRISMA flow diagram

Authors:

Elizabeth Stovold (estovold@sgul.ac.uk)
Deirdre Beecher (Deirdre.Beecher@lshtm.ac.uk)
Ruth Foxlee (RFoxlee@cochrane.org)
Anna Noel-Storr (anna.noel-storr@rdm.ox.ac.uk)

Version: 2 Date: 16 May 2014

Author's response to reviews:

Please see covering letter